Author's response to reviews

Title: Student centered curricular elements are associated with healthier educational environment and lower depressive symptoms in medical students

Authors:

Eiad Abdelmohsen AlFaris (efarisx2@yahoo.co.uk)
Naghma Naeem (naghma.naeem@gmail.com)
Farhana Irfan (masoodfarhana1969@gmail.com)
Riaz Qureshi Qureshi (riaz.qureshi9@gmail.com)
Cees van der Vleuten (c.vandervleuten@maastrichtuniversity.nl)

Version: 2 Date: 10 September 2013

Author's response to reviews:

A statement about Institutional Review Board approval has been added in the Methods section of the manuscript, including the name of the body which gave approval and reference number.

A section about no competing interests, financial support and author's contribution has been added after the conclusion in the manuscript.